Author's response to reviews

Title: The CAP cancer protocols - A case study of caCORE based data standards implementation to integrate with the Cancer Biomedical Informatics Grid

Authors:

Jonathan Tobias (jotobias@mednet.ucla.edu)
Ram Chilukuri (ram.chilukuri@semanticbits.com)
George A Komatsoulis (komatsog@mail.nih.gov)
Sambit Mohanty (mohantys2@upmc.edu)
Nicholas Sioutos (nsioutos@mail.nih.gov)
Denise B Warzel (warzeld@mail.nih.gov)
Lawrence W Wright (lwright@mail.nih.gov)
Rebecca S Crowley (crowleysrs@upmc.edu)

Version: 6 Date: 9 June 2006

Author's response to reviews:

The abstract was changed to match the requested structure of Background, Methods, Results, Conclusions. The Abstract details were also updated on the submission page.

The competing interest statement was written as "The authors declare that they have no competing interests."

Reference 10 was removed from the reference list and referred to as "unpublished findings" in the body of the manuscript.

The figures were cropped to minimize white space.

Reviewer's comments: page 31. The text was revised to clarify MVR as MissingValueReason pages 26-27. The text was changed to make the example in Figure 3 match the text for consistency.